Author's response to reviews

Title: Genome-wide expression patterns associated with oncogenesis and sarcomatous transdifferentation of cholangiocarcinoma

Authors:

Min&A Seol (minna001@hanmail.net)
In&Sun Chu (chu@kribb.re.kr)
Mee&Jin Lee (wkdb82@hanmail.net)
Goung&Ran Yu (senayu@hanmail.net)
Xiant&Dan Cui (xj-cui@hanmail.net)
Baik&Hwan Cho (chobh@chonbuk.ac.kr)
Eun-Kyung Ahn (mercury0916@hanmail.net)
Sun-Hee Leem (shleem@dau.ac.kr)
In-Hee Kim (ihkimmd@chonbuk.ac.kr)
Dae-Ghon Kim (daeghon@chonbuk.ac.kr)

Version: 8 Date: 6 February 2011

Author's response to reviews:

Major revisions
1. The talbes are reformatted.
2. The reference (41) is corrected.

Minor revisions
1. Email addresses are rechecked.